CS340 Winter 1996 Ming Li

CS 340 Data Structures and Algorithms, Winter 1996

Section 1 11:30 AM – 12:30 MWF, MC 4058 Section 2 2:30 PM – 3:30 PM MWF, MC 4058 Instructor Ming Li, office: DC 2339; phone: 888-4659; e-mail: mli@math; Office Hours:

3:30-5:00 PM MW, or by appointment or just stop by.

Teaching Assistants ffl Karsten Verbeurgt, DC3145, x3497, kaverbeurgt@neumann ffl Ka Yee Yeung, DC 2127, x3299, kyyeung@neumann ffl Hao Zhang, DC3529, x3590, hzhang@neumann Textbook Data Structure and Their Algorithms by Harry Lewis and Larry Denenberg, HarperCollinsPublishers, 1991. It is available in the university bookstore. Lecture notes (overheads) will be gradually made available, as the course goes on, in the cs340 homepage: http://www.math.uwaterloo.ca/~mli/cs340.html.

Computing Accounts Most students will already have accounts on undergrad.math. Extra quota is added to your account for CS340.

You should do your programming assignments using either the languages Pascal or C (but not C++).

There is a news group called uw.cs.cs340 for questions, discussions, and announcements. We will also extensively use our web page to provide course information, lecture notes, and handout materials.

Marking Scheme ffl Four assignments handed out on

Jan 10, due Jan 24, worth 10 marks; Jan 26, due Feb 9, worth 10 marks; Feb 21, due Mar 6, worth 10 marks; Mar 8, due Mar 22, worth 10 marks.

ffl A midterm exam (temporarily) on Wednesday, Feb 14, 6-8 PM, worth 20 marks.

ffl A final exam worth 40 marks.

Handing in the Assignments Please deposit the assignment on the due date in the slot marked CS340 (with correct section number) of the black assignment cabinets on the 3rd floor of the MC building. Late assignments will be accepted for half credit within 3 days of the due date. No later assignments will be accepted after that.

Do Not Copy Assignments Discussion with others is encouraged but copying other’s assignments is forbidden. If copying or other forms of cheating are detected, the penalty is -100% for the work, with a minimum loss of 5% of the course mark.